

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Xu, Yue; Liu, Yunguo; Lobo, Gerald J.

# **Article**

Troubled by unequal pay rather than low pay: The incentive effects of a top management team pay gap

China Journal of Accounting Research

# **Provided in Cooperation with:**

Sun Yat-sen University

Suggested Citation: Xu, Yue; Liu, Yunguo; Lobo, Gerald J. (2016): Troubled by unequal pay rather than low pay: The incentive effects of a top management team pay gap, China Journal of Accounting Research, ISSN 1755-3091, Elsevier, Amsterdam, Vol. 9, Iss. 2, pp. 115-135, https://doi.org/10.1016/j.cjar.2016.01.001

This Version is available at: https://hdl.handle.net/10419/187652

# Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

# Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



https://creativecommons.org/licenses/by-nc-nd/4.0/





Contents lists available at ScienceDirect

# China Journal of Accounting Research

journal homepage: www.elsevier.com/locate/cjar



# Troubled by unequal pay rather than low pay: The incentive effects of a top management team pay gap \*\*,\*\*\*\*



Yue Xu<sup>a,1</sup>, Yunguo Liu<sup>a,\*</sup>, Gerald J. Lobo<sup>b</sup>

#### ARTICLE INFO

# Article history: Received 7 March 2015 Accepted 12 January 2016 Available online 15 February 2016

JEL classification:

D230 D820

J310

Keywords: TMT pay level Pay gap Property rights SOE salary reform

#### ABSTRACT

We examine the relationships with firm performance of the internal pay gap among individual members of the top management team (TMT) and the compensation level of TMT members relative to their industry peers. We find that pay gap is positively related to firm performance and that this positive relation is stronger when the TMT pay level is higher than the industry median. However, we do not observe such effects in Chinese state-owned enterprises (SOEs), in which both the executive managerial market and compensation are government-regulated. We also document that cutting central SOE managers' pay level can increase firm value, whereas doing so for local SOE managers has the opposite effect. Our findings have important implications for research on TMT compensation as well as for policy makers considering SOE compensation reform.

© 2016 Sun Yat-sen University. Production and hosting by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

#### 1. Introduction

Top management team (TMT) incentive-based contract design is a topic of considerable interest to academics and practitioners. It is an especially important issue in China because both the level of TMT

<sup>&</sup>lt;sup>a</sup> School of Business, Sun Yat-sen University, China

<sup>&</sup>lt;sup>b</sup> C.T. Bauer College of Business, University of Houston, United States

<sup>\*</sup> We acknowledge the support of the National Natural Science Foundation of China (71372150, 71572197, and 71032006).

<sup>\*\*</sup> We acknowledge the helpful comments and suggestions provided by Professor Bin Miao from the National University of Singapore, the anonymous reviewers, and participants at the CJAR special issue symposium in Suzhou in April 2015.

<sup>\*</sup> Corresponding author. Mobile: +86 186 6608 0107.

E-mail addresses: xinyan1991@126.com (Y. Xu), sysulyg@gmail.com (Y. Liu), gjlobo@uh.edu (G.J. Lobo).

<sup>&</sup>lt;sup>1</sup> Mobile: +86 135 7030 8719.

compensation has increased considerably and the compensation differentials among members of the TMT have widened significantly following the introduction of market-oriented reforms in China in 1978. Our focus in this study is on implications of the variation and the level of TMT pay for firm performance.

We define the variation in pay among the TMT as the difference between the CEO's pay and that of the other executives in the TMT, and refer to this difference as "TMT pay gap." TMT pay gap thus concerns the reward that non-CEO executives can expect if they are promoted to CEO. Prior research makes conflicting predictions about whether a large TMT pay gap promotes competition among TMT members and whether it enhances firm performance. Tournament theory, on the one hand, posits that a pay gap among different organizational levels provides a competition incentive such that the larger the pay gap, the better the firm's performance. Social comparison theory, on the other hand, stresses teamwork and cooperation within the TMT, and thus posits that a smaller pay gap can improve satisfaction and willingness to cooperate, thereby boosting firm performance. However, whether a team engages in competition or cooperation is an empirical issue (Harbring and Irlenbusch, 2003), and this relationship is moderated by many factors, including task dependence and the individual incentive system (Shaw et al., 2002; Kepes et al., 2009).

In addition to TMT pay gap, the level of TMT pay is also an important factor that affects firm performance. We define "TMT pay level" as the difference between the average pay of the TMT and the average pay of industry peer TMTs (Gerhart and Milkovich, 1992), and refer to this difference as "TMT pay level." TMT pay level thus reflects the external competitiveness of the firm's compensation policy (He and Hao, 2014). TMT pay level may have both a direct and an indirect effect on firm performance. First, as an important factor in the TMT's incentive system, TMT pay level has direct implications for firm performance. Second, TMT pay level may also have an indirect effect on firm performance because it moderates the relationship between TMT pay gap and firm performance. Excluding the effects of pay level from the model would thus result in a biased estimate of the relationship between TMT pay gap and firm performance. However, most of the related research in China does not consider this interactive effect of TMT pay level (Lin et al., 2003; Chen and Zhang, 2006). Additionally, firms often use industry peers as a benchmark when negotiating contracts with top executives (Jiang, 2011). Different TMT pay levels lead to an external comparison between firms, and top executives and then form corresponding levels of satisfaction with their compensation, which in turn influences the competition–cooperation relationship within the TMT.

As indicated earlier, we conduct our empirical analysis using compensation data from Chinese firms. The market-oriented reforms introduced in China in 1978 have led to substantial increases in the compensation levels of some executives, especially those at monopoly and public welfare firms. Further, the pay gap among Chinese firms' TMTs has considerably widened during this period (Zhang, 2008; Li and Hu, 2012). Concerned about the widening pay gap and increasing compensation level, China's Central Political Bureau passed a resolution on 29 August 2014 to reform the pay system for the responsible persons of centrally managed companies. The program focuses on five main areas: (1) improvement of the reward system, (2) adjustment of the pay structure, (3) strengthening of supervision, (4) regulation of the pay level, and (5) treatment standardization. The last two areas, in particular, are intended to address unreasonably high incomes and pay gaps to promote social justice.

"Pay gap" can refer either to the income gap between executives and general staff or the gap between TMT members' pay. Chinese are generally more sensitive to the former gap, particularly since the round of pay cuts and layoffs in 2008 that saw executives retain high pay levels (Liu and Sun, 2010). However, because the TMT is at the highest managerial level of the firm, the within-team pay gap is related to the distribution of limited compensation among executives, and thus plays an important role in the TMT incentive system. Moreover, if the overall TMT pay level is adjusted, the question is whether and how income should be distributed among team members to ensure the effectiveness of the compensation incentive mechanism. To answer this question, we explore the relationship between a TMT's overall pay level and the pay gap among its members.

Since 2005, it has been mandatory for China's listed firms to disclose their executives' compensation. In this study, we examine whether TMT pay level affects the relationship between a TMT pay gap and firm

<sup>&</sup>lt;sup>2</sup> For example, Knoeber and Thurman (1994) point out that Ehrenberg and Bognanno (1990a,b) ignore the incentive effect of prize level when using the behavior of professional golfers to examine tournament theory. Because the prize structure was identical across tournaments, larger prize gaps were always the result of higher prize levels.

performance based on executive compensation data from 2005 to 2012. Moreover, in the context of the current reform of China's state-owned enterprises (SOEs), we further analyze the relationship between TMT pay level and pay gap incentive efficiency under different ownership conditions. We find that TMT pay level can moderate pay gap incentive efficiency and that a pay gap within a TMT exerts stronger tournament effects when TMT members' pay level is higher than that of their industry peers. In Chinese SOEs, however, because both the market for executives and their compensation are regulated by the government, even a high pay level fails to stimulate tournament effects in the TMT. In central SOEs, cutting managers' pay level can increase the value of the firm. However, doing so in local SOEs is likely to dampen tournament incentive efficiency. For local SOEs that provide high TMT pay level, a larger TMT pay gap is associated with better firm performance.

This study makes several contributions. First, it enriches the literature on TMT pay incentives. These incentives comprise the entire team's pay level (the first distribution) and its internal pay gap (the second distribution) (Zhang et al., 2012). Many studies discuss these two aspects of incentives separately, but few combine them. Our findings show that the TMT pay level influences the team's internal distribution efficiency. When the pay level is higher than that of industry peers, a larger pay gap induces better tournament incentives and performance. However, when the pay level is lower than that of industry peers, a larger pay gap may be harmful to firm performance.

Second, our study adds to the literature on pay gaps. Tournament theory and social comparison theory do not coincide in terms of their predictions of the pay gap incentive effect. We argue that the applicability of the two theories depends on external equity. When top executives perceive external fairness, tournament theory is more applicable; when they perceive less external fairness, social comparison theory is more applicable. This finding also supplements Brown et al. (2003).

Third, our study contributes to the literature on Chinese SOE reform. Wu et al. (2010) and Li et al. (2014) report that excess compensation does not have incentive effects. Similarly, our findings show that excess pay does not promote internal competition in the TMT. For central SOEs, reducing top managers' pay level can stimulate managerial competition and enhance firm value, which lends support to the recently passed and implemented decision to reduce the compensation of central SOEs' top executives. At the same time, our results suggest that the reform policy for local SOEs should not simply follow than that of central SOEs. Reducing pay levels may not be the optimal decision for local SOEs, although adjusting the internal pay structure may enhance firm performance.

The remainder of the paper is organized as follows. Section 2 presents the literature review and hypothesis development, and Section 3 discusses the research design. Section 4 reports the descriptive statistics and the results of the empirical analyses, while Section 5 discusses the results of robustness tests. Section 6 presents the results of additional analyses and Section 7 concludes the paper.

# 2. Literature review and hypothesis development

# 2.1. Literature on pay gap and pay level

There is a sizeable body of research on the relationship between TMT pay gap and firm performance. One strand of research is tournament theory, developed by Lazear and Rosen (1981), which posits that although a non-CEO's salary may double within a day following the promotion to CEO, it would be difficult to argue that his/her ability has also doubled within a day. Thus, it is difficult to explain the pay gap in TMTs through recourse to traditional economic theory, which argues that pay levels are determined by marginal output (e.g., Lin et al., 2003). Tournament theory argues that under the conditions of cooperative effort and task interdependence, it is not feasible to set executives' pay based on their marginal output when monitoring is difficult. Although marginal output-based pay for executives might seem more equitable, a large pay gap can encourage rank-order competition, thereby improving firm performance. Numerous studies provide empirical evidence consistent with the predictions of tournament theory. For example, Ehrenberg and Bognanno (1990a, 1990b) study the behavior of professional golf players, and find that increases in the differential between prizes provide them with an incentive to exert more effort. Main et al. (1993) report a positive relationship between pay gap and return on assets for a sample of more than 200 firms and 2000 executives per year over a five-year

period. Rosen (1986), Eriksson (1999), Lambert et al. (1993) and Kale et al. (2009) also provide theoretical and empirical evidence of the theory's efficacy.

Another strand of research is social comparison theory, which, in contrast to tournament theory, argues that people pay close attention to equity in most situations. Particularly when a team is working jointly toward the same objective, team members are likely to not only being concerned about their own income, but also to compare it with those of their fellow team members to judge whether the income distribution is fair. If a TMT's internal pay gap is sufficiently large to seem inequitable to non-CEO executives, dissatisfaction with the income distribution may reduce their willingness to cooperate, thus damaging firm performance. Therefore, social comparison theory places greater stress on the importance of a compressed pay distribution in TMTs (Lazear, 1989; Pfeffer and Langton, 1993; Cowherd and Levine, 1992). O'Reilly et al. (1988) suggest that a compressed pay distribution and smaller pay gap encourage collaboration among employees, and are thus beneficial to firm performance. Drago and Garvey (1998), Bloom (1999), Hibbs and Locking (2000) and Fredrickson et al. (2010) also provide empirical evidence in support of social comparison theory.

The two theories make contradictory predictions about the relationship between a TMT pay gap and firm performance. Tournament theory argues that a larger TMT pay gap provides an incentive for executive competition, whereas social comparison theory posits that a larger TMT pay gap causes non-CEO executives to feel deprived and is not conducive to cooperation. In reality, most members of organizations, particularly top executives, work toward a common objective and require interdependence or collaboration. Cooperation and competition coexist among executives in a TMT, and we thus need empirical evidence to determine which theory is dominant in a specific setting. For example, Lin et al. (2003) propose that in Chinese listed firms, tournament incentives' positive effect exceeds the negative effect brought about by feelings of unfairness and non-cooperative behavior. They thus conclude that tournament theory is more suitable for explaining the relationship between TMT pay gaps and firm performance in Chinese firms.

However, the literature in this area is not limited to studies seeking supportive empirical evidence for these two theories. Scholars have discussed the aforementioned relationship in a variety of contexts (Trevor and Wazeter, 2006). Siegel and Hambrick (2005) examine the interactive effect of technological intensiveness and TMT pay gaps on firm performance, finding that in high-technology firms that require greater technological intensiveness, pay gaps are detrimental to firm performance. Kepes et al. (2009) report a positive relationship between pay gaps and firm performance when those gaps are attributable to the use of performance-based pay because employees feel that the distribution process is fair. Brown et al. (2003) use a large database of hospitals to examine the modulating effect of an organization's pay level on internal pay gap incentive efficiency.

As noted, "pay level" in this paper refers to top executives' average pay level relative to that of their industry peers (Gerhart and Milkovich, 1992). Milkovich and Newman (2002) describe pay levels as leading, matching or lagging the market. The traditional economic literature primarily uses efficiency wage theory, proposed by Akerlof and Yellen (1986), to explain why a firm's pay level affects its performance. If the firm's pay level exceeds that of its peers, it will find it easier to attract, retain and motivate outstanding talent. The Chinese literature explains the influence of pay level on firm performance primarily from the perspective of external equity. Wu et al. (2010) estimate the excess compensation of top managers as a measure of external equity, and find that high pay levels motivate non-SOE managers, but have no effect on SOE top managers. Qi and Zou (2014) use relative quantiles of top executives' average pay as a measure of external pay equity, reporting that a higher pay level always increases managers' perceptions of fairness, and thus motivates them to exert greater effort to boost firm performance. From the perspective of the managerial market, Li et al. (2014) demonstrate that only when a TMT's pay level leads the market does excess compensation for top managers provide positive incentives. Therefore, the TMT's pay level offers the same incentives to top managers as an internal TMT pay gap. Accordingly, in order to focus on the incentives of the latter, we must exclude the direct and indirect effects of pay level. In other words, a study on the effects of pay gaps on firm performance must control for pay level.

The Chinese literature on pay gaps considers both the gap between management and workers and that among managers within TMTs. The Chinese tend to be more sensitive to the former because of the traditional pull of the harmonious society principle, and thus the pay distribution within TMTs has received little attention. Interestingly, however, almost all existing papers on the consequences of an internal TMT pay gap are

consistent in supporting tournament theory (Lin et al., 2003; Chen and Zhang, 2006), although they are somewhat limited by their failure to control for pay level. Zhang (2007) examines the relationship between within-TMT pay gaps and firm performance while holding pay level constant, and finds a negative relationship, which is consistent with the prediction of social comparison theory. This result suggests that, for Chinese listed firms, the TMT pay level may be an influential factor in TMT pay gap efficiency.

At present, the Chinese literature on TMT pay level and pay gap is fragmented. Many studies discuss one or the other in isolation, but few consider them together. In a field survey of 376 general workers from 59 departments of Chinese firms, He and Hao (2014) find that within-department pay differences exert a negative effect on workers' emotional commitment only when the overall departmental pay level is lower than that of other departments. In this study, we combine TMT pay levels with a study of TMT pay gap efficiency, in an attempt to contribute to research on top management compensation mechanisms in Chinese listed firms.

# 2.2. Hypothesis development

As previously discussed, tournament theory emphasizes competition within the organization, whereas social comparison theory stresses cooperation and fairness within the organization. Colquitt et al. (2001) suggest that perceived equity is the result of a subjective judgment process and that subjectivity is reflected primarily in the choice of reference object. Many studies propose that employees compare their salaries with several reference objects (Brown, 2001; Hills, 1980; Law and Wong, 1998). Oldman et al. (1986) divide the comparison with reference objects into internal organizational justice and external organizational justice. A within-TMT pay gap influences internal organizational justice, whereas TMT pay level influences external organizational justice. These two kinds of justice are not independent, but rather interact with each other (Trevor and Wazeter, 2006). It has been mandatory for Chinese listed firms to disclose their executives' compensation since 2005, which makes it easier for top executives to obtain payment information on their peers and to engage in external comparison.

Lazear and Rosen (1981) propose a two-player tournament model in which the players' effort and performance levels are positively related to the size of the reward, but have nothing to do with the prize level. Their model assumes the prize level to be exogenously given, which limits the possibility of players making external comparisons. However, this assumption does not hold for workers' wages in most firms. Efficiency wage theory (Akerlof and Yellen, 1986) states that providing employees with a market-leading wage is beneficial to the firm because employees are more willing to remain in a high-paying organization. Messersmith et al. (2011) find that top executive turnover is less likely when executives receive a higher proportion of overall TMT compensation. Not only is original talent retained, but other outstanding managers are also attracted to the team. Therefore, when there is a within-TMT pay gap, a high TMT pay level exerts a natural incentive effect on the CEO and, more importantly, the negative effect on non-CEOs arising from the internal pay gap is partially offset. Bloom and Michel (2002) suggest that if a firm's payment level is higher than the market's, many of the negative consequences of an internal pay gap are alleviated. Frank (1985) proposes that it is easier for employees to accept an unfair wage distribution when their wages are higher than their marginal output. Trevor and Wazeter (2006) find that when employees are in a lower pay position internally, a higher pay position externally enhances their perception of pay equity. Hence, an internal pay gap is less likely to exert negative effects and more likely to create tournament incentives.

However, if the TMT's pay level lags the market, all of the team's top executives are likely to perceive a low degree of external equity, that is, to perceive themselves at a disadvantage relative to their industry peers. Although a large internal pay gap may somewhat alleviate the CEO's dissatisfaction, it will worsen that of non-CEOs, who are in a poorer pay position relative to both internal and external referents. Such double discontent with their compensation may well reduce their willingness to cooperate (Deutsch, 1985; Pfeffer and Langton, 1993) or even encourage them to desert the firm (Bloom and Michel, 2002). Both outcomes

<sup>&</sup>lt;sup>3</sup> Chen et al. (2011) also control for the average pay level of the three top executives other than the CEO in their model. Their results are consistent with tournament theory. However, even when these top three executives' pay is controlled for, a larger pay gap always accompanies higher CEO pay, and thus it is still difficult to distinguish whether the positive relationship between a pay gap and firm performance stems from high CEO pay or a large pay gap.

are potentially damaging to firm performance. Moreover, if the managerial market is effective, those who feel underpaid will move to other companies, where they can obtain ability-matched compensation, and those who choose to remain on a poorly paid team may be incompetent and thus struggle to find a better paid job. Messersmith et al. (2011) provide evidence to show that reducing overall TMT compensation increases executive turnover. Therefore, a low TMT pay level and large within-team pay gap induce a number of negative effects, such as a lack of cooperation and higher turnover among non-CEOs, thereby offsetting the positive tournament incentives of the pay gap to a large degree. It is possible that the negative effect of an internal pay gap is dominant when overall pay is below a certain level.

The preceding discussion suggests that a TMT pay gap can produce positive tournament incentives, especially when the TMT pay level is above than that of peer firms. It also suggests that a TMT pay gap can produce negative effects, especially when external equity perceptions brought about by comparison with other TMT pay levels influence internal equity perceptions induced by the pay gap. Although it is unclear whether the positive tournament effects or the negative effects of internal inequities dominate, it is clear from the above discussion that the positive tournament effects of the TMT pay gap will be more positive when the TMT pay level is above than that of peer firms, and the negative effects are more negative when the TMT pay level is below than that of peer firms. This implies that the difference in the relationship between the TMT pay gap and performance will be positive for higher versus lower TMT pay levels. Therefore, we formulate our first hypothesis as follows:

**H1.** The relationship between TMT pay gap and firm performance is likely to be more positive when TMT pay level is higher than that of peer firms.

Although there is considerable evidence supporting the predictions of tournament theory, its efficacy in China is unclear because of the unique features of China's institutional environment that considerably differ from those of mature market economies. For example, China has many SOEs whose top managers earn higher average pay levels than their non-SOE counterparts, and the TMT pay gap is significantly lower in SOEs than in non-SOEs. The primary reason for this discrepancy is that the compensation of SOE top managers is regulated by the Chinese government.

Given these differences, a question of interest is whether tournament theory is capable of explaining the incentives of top managers in Chinese SOEs. Chinese scholars (e.g., Zhou and Zhu, 2010) suggest that the tournament incentive mechanism is actually encouraged by China's institutional environment. Because the government is the ownership representative of the people, it has a natural information disadvantage. Also, SOE managers are multitaskers. It is thus difficult for the government to find appropriate measures to evaluate SOE managers, and tournament promotion based on relative performance is a widely used measure. Moreover, competence for the CEO position is also judged by political promotion, as most SOE managers are appointed by the government and also have an administrative ranking within the government. Although China has attempted in recent years to implement non-administrative SOE reform, many top managers still treat promotion within the SOE as a way to further their political careers. Yang et al. (2013) study the "quasi-official" promotion mechanism in central SOEs. They assess whether internal promotion within an SOE's TMT is more political than that within a non-SOE's TMT and whether the CEOs of SOEs are able to obtain more political profits, such as political rent-seeking opportunities and transfers to government departments. Therefore, even a very small within-TMT pay gap in an SOE can create tournament incentives and motivate managers.

Of course, under China's tradition of egalitarian thought, the requirement for a fair income distribution may offset the positive effects of such a pay gap to some extent. However, we are not concerned with predicting the direction of a pay gap's net effects, but rather with determining how the TMT pay level, relative to the market or to industry peers, affects TMT pay gap incentives in firms with different ownership types.

First, the compensation that SOE top managers receive is government-regulated and separated from market conditions. Thus, their disclosed pay level is not necessarily indicative of the competitiveness of their

<sup>&</sup>lt;sup>4</sup> It should be noted that political profits here do not equate to non-monetary income. Non-monetary income refers to such benefits as reputation and status after promotion to CEO, whereas political profits refer specifically to the benefits to the executive's post-promotion political career.

payment capacity. Second, because monetary compensation is regulated, managerial perks have become an alternative means of compensating SOE managers (Chen et al., 2005). Thus, these managers' disclosed compensation does not fully reflect their incentives. They have more implicit incentives that are not sensitive to pay comparisons with peers. For example, Li et al. (2014) suggest that even if SOE managers obtain a higher pay level than their industry peers, excess payment does not bring any added incentives. However, if SOE managers, most of whom are appointed by the government, obtain a lower pay level than their peers, even if they also have an external perception of unfairness, they are unlikely to slack off at work or desert their positions because they are concerned about their careers in the state-owned system. In summary, the TMT pay level in SOEs has little effect on the input and output of top managers.

As previously noted, the premise that the TMT pay level moderates the effects of a TMT pay gap on firm performance is based on the assumption that top managers compare themselves with external referents, thereby exacerbating the perception of internal injustice induced by the pay gap. In non-SOEs, compensation information on top managers is relatively more transparent and focused on monetary incentives (Lu et al., 2012), and the median values of pay in the same industry are more often used as a benchmark (Jiang, 2011). Hence, non-SOE managers are more concerned with peer comparisons than their SOE counterparts. Moreover, because of the relatively high marketization level of non-SOE managers, professional managers can flow freely in the market, and can thus choose job-hopping, slacking off at work or deserting their position in response to a pay gap-induced perception of injustice, which in turn damages firm performance. SOE managers, in contrast, are not sensitive to the competitiveness of their pay level, and thus external compensation comparisons have little effect on internal TMT competition and cooperation. This discussion leads to our second hypothesis:

**H2.** The effect of TMT pay level on the relationship between TMT pay gap and firm performance will be more pronounced for non-SOEs than for SOEs.

# 3. Research design

#### 3.1. Data and sample selection

Executive compensation data and other firm characteristic data are sourced from the China Stock Market & Accounting Research (CSMAR) database. It has been mandatory for Chinese listed firms to disclose their executives' compensation details since 2005. Accordingly, our sample period spans 2005–2012, and the sample contains all Chinese listed firms with available data in that period. We define TMT members as all those who occupy a management position. Members of boards of directors and supervisors, who receive only a fixed bonus, are excluded. The top managers in our sample include general managers, deputy general managers, chief engineers, chief accountants, and chief financial officers. The general manager, president or CEO of the firm are defined as the CEO (Liao et al., 2009), and all other managers as non-CEOs. After collecting executive compensation data from the CSMAR database, we screened the data manually according to position titles. The screening steps were as follows: (1) obtain the *comprehensive management file* and the *executive individual compensation file* from the CSMAR database and check the CEO compensation data; (2) drop observations if the CEO changed in the current year<sup>6</sup>; and (3) drop observations if CEO compensation was not disclosed or the CEO received only a fixed bonus. After screening, we had 11,589 valid CEO observations. We then screened non-CEO compensation data from the *executive individual compensation file* by the aforementioned position titles.

After removing (1) 163 financial companies, (2) 566 companies for which we could not calculate the TMT pay gap because CEO compensation was lower than the median compensation for non-CEOs, (3) 193

<sup>&</sup>lt;sup>5</sup> The titles of the chief managers in Chinese listed firms are not consistent, with some called general manager and some called president. In this paper, we uniformly refer to them as CEOs.

<sup>&</sup>lt;sup>6</sup> If the CEO changed in the current year, then the reported CEO compensation in that year did not constitute data for a whole year, possibly leading to large deviations in calculating pay gap.

<sup>&</sup>lt;sup>7</sup> As in Liao et al. (2009), this condition might arise if the CEO receives only part of his or her compensation from this firm or if specialists who are entitled to higher compensation are hired as non-CEOs.

observations whose debt to asset ratio was greater than 1, (4) 1197 observations with missing values, <sup>8</sup> and (5) 284 observations that received special treatment, we were left with 9186 valid firm-level observations for our empirical analysis.

# 3.2. Variable definitions and model specification

In accordance with the foregoing theoretical analysis and relevant research (Kale et al., 2009; Chen et al., 2011; Zhang, 2007), we use the following model (1) to examine how a firm's TMT pay level influences the relation between an internal TMT pay gap on firm performance:

$$PERF_{t} = \beta_{0} + \beta_{1}GAP_{t} + \beta_{2}PL_{t} + \beta_{3}GAP_{t} \times PL_{t} + \beta_{4}SOE_{t} + \beta_{5}LEV_{t} + \beta_{6}SGROW_{t} + \beta_{7}SIZE_{t}$$

$$+ \beta_{8}TOP1_{t} + \beta_{9}CEOAge_{t} + \beta_{10}BOARD_{t} + \beta_{11}INDEPB_{t} + \beta_{12}DUAL_{t} + \beta_{13}MKR_{t}$$

$$+ \beta_{14}Option_{t} + \sum Industry_{t} + \sum Year_{t} + \varepsilon_{t}$$

$$(1)$$

where PERF is either a measure of accounting performance (ROA) or market performance (TOBINQ).

GAP refers to the TMT pay gap, our main variable of interest. Following Bognanno (2001), Kale et al. (2009), Chen et al. (2011) and Kini and Williams (2012), we define GAP as the reward received after a non-CEO executive is promoted to CEO. We use two measures of GAP. The first measure,  $GAP_{\_}I = ln$  (CEO pay-median pay of non-CEOs), where "ln" refers to the natural logarithm, and the second measure,  $GAP_{\_}2 = CEO$  pay/median pay of non-CEOs, in accordance with Lin et al. (2003), Zhang (2008) and Hambrick and Siegel (1997).

*PL* measures the TMT average pay level relative to the median of the average pay levels of all other firms in the same industry. It represents the external competitiveness of the TMT's pay. *PL* equals 1 if the firm's average TMT pay is higher than the yearly median level of firms in the same industry, and 0 otherwise.

The other control variables, including firm characteristics and firm governance, are defined as in the literature (Chen et al., 2011; Lin and Lu, 2009). For example, an SOE (ownership status) value of 1 indicates that the firm is state-owned, whereas an SOE value of 0 indicates it is not. Xu et al. (2006) suggest that whether the controlling shareholder is state-owned or non-state-owned has a significant influence on firm performance. The liability-asset ratio (LEV) is equal to total liabilities divided by total assets. Myers (1977) and Jensen (1986) both find LEV to affect firm performance, although the former reports a negative effect and the latter a positive effect. Sales growth (SGROW) represents the firm's rate of growth, and is equal to the difference between current-year sales and previous-year sales divided by previous-year sales. The other control variables are defined in Table 1. We also control for industry and year fixed effects.

Endogeneity is a serious concern when studying the relationship between pay gap and firm performance. Li and Hu (2012) suggest that there is a natural positive relationship between these two variables in Chinese SOEs. Kale et al. (2009), Chen et al. (2011), Lin and Lu (2009) and Kini and Williams (2012) all adopt instrumental variable (IV) estimation or two-stage least-squares (2SLS) estimation to control for the endogenous relationship between pay gap and firm performance. We choose lagged TMT pay gap (*LGAP*) and median TMT pay gap in the same industry (*MedianGAP*) as the IVs for *GAP*. Kale et al. (2009) suggest that TMT pay gap is positively associated with the median *GAP* of industry peers.

There is also concern about endogeneity in the relationship between TMT pay level and firm performance, as firms with a higher pay level are likely to have better performance. To address this concern, we follow Fang (2012) and Wu et al. (2010) and estimate a TMT's excess pay level after removing the effects of firm performance, firm characteristics and other factors. We use the following model (2) to estimate the excess pay level (Core et al., 1999; Fang, 2012).

<sup>&</sup>lt;sup>8</sup> Missing-value observations stem primarily from our need for data from two consecutive years. If a firm was listed in the sample period, the listing year is that firm's first year, and no data are available for the previous year.

<sup>&</sup>lt;sup>9</sup> The results are similar when we use the mean of non-CEO pay, instead of the median.

Table 1 Variable definitions.

Name	Variable	Definition
Firm Performance	ROA TOBINQ	Net income/ending total assets (Market value of tradable shares + net assets + market value of net debt)/ending total assets
TMT Pay Gap	GAP_1 GAP_2 LGAP MedianGAP	In (CEO pay-median pay of non-CEOs) CEO pay/median pay of non-CEOs Lagged GAP (GAP_1 or GAP_2) Median GAP (GAP_1 or GAP_2) value of each industry in each year
TMT Pay Level	PL UF	Dummy variable that equals 1 if firm <i>i</i> 's average TMT pay is higher than the median TMT pay of firms in the same industry each year, and 0 otherwise Excess pay of the TMT, estimated as the residuals of model (2)
Firm/Executive Characteristics	SOE  LEV SGROW  SIZE CEOAge Option	Ownership status, a dummy variable that equals 1 if the firm is state-owned, and 0 otherwise  Total debt/total assets  Sales growth rate = (sales revenue of current year – sales revenue of previous year)/ sales revenue of previous year  In (ending total assets)  In (CEO age)  Dummy variable that equals 1 if the top executives have been granted stock options, and 0 otherwise
Firm Governance	TOPI BOARD OUT DUAL MKR	Percentage of outstanding shares held by the firm's largest shareholder Number of directors on the firm's board Number of independent directors/number of directors of the board Dummy variable that equals 1 if the CEO is also the chairperson, and 0 otherwise Based on the Chinese regional market index estimated by Fan and Wang (2007), <i>MKR</i> is a dummy variable that equals 1 if the firm is in a region with a market index higher than the median value, and 0 otherwise

$$LnCOMP_{t} = \beta_{0} + \beta_{1}SIZE + \beta_{2}LEV_{t} + \beta_{3}ROA_{1} + \beta_{4}ROA_{t-1} + \beta_{5}DUAL_{t} + \beta_{6}BOARD_{t} + \beta_{7}SOE_{t}$$

$$+ \beta_{8}MHOLD_{t} + \beta_{9}BM_{t} + \varepsilon_{t}$$
(2)

where  $LnCOMP_t$  is the natural logarithm of the TMT's average pay,  $ROA_{t-1}$  is lagged firm performance, MHOLD is the TMT's average shareholding ratio, BM is the ratio of book value to market value of equity, and the other variables are as defined in Table 1.

We first estimate model (2) for each industry-year and use the residuals from this model as our estimate of excess pay of the TMT. We also use this excess pay measure as the IV estimator for TMT pay level.

Returning to model (1), the interaction between TMT pay level and TMT pay gap may also suffer from an endogeneity problem. If we use the IVs for GAP and PL, the interaction term  $GAP \times PL$  should also use the interaction of these IVs. Therefore, we use four IVs, namely LGAP, MedianGAP, MedianG

According to H1, if a higher TMT pay level is accompanied by better TMT pay gap-induced tournament incentives, the regression coefficient on  $GAP \times PL$  should be positive and significant, <sup>11</sup> and, according to H2, that positive relationship should be stronger in the non-SOE subsample.

<sup>&</sup>lt;sup>10</sup> These tests include unidentifiable inspection, the weak identification test and the Sargan test.

<sup>&</sup>lt;sup>11</sup> Because a TMT pay gap may have positive or negative effects, we do not predict *GAP*'s direction in model (1) as Siegel and Hambrick (2005) do.

Table 2 Descriptive statistics for main variables.

Variable	Number of observations	Mean	Minimum	Median	Maximum	Standard deviation
ROA	9186	0.040	-0.389	0.037	0.208	0.057
TOBINQ	9186	1.727	0.714	1.370	7.222	1.083
$PL\_DIF$	9186	175.098	36.825	149.563	564.233	112.872
$PAY\_GAP$	9186	18.646	0.500	10.847	150.466	24.216
$GAP\_1$	9186	11.552	8.476	11.594	14.366	1.123
$GAP\_2$	9186	1.642	1.000	1.429	4.814	0.658
PL	9186	0.527	0	1	1	0.499
SOE	9186	0.561	0	1	1	0.496
LEV	9186	0.474	0.050	0.489	1.000	0.204
SGROW	9186	0.211	-0.778	0.144	4.317	0.480
SIZE	9186	21.667	18.698	21.522	25.620	1.180
CEOAge	9186	3.853	3.219	3.850	4.317	0.133
TOP1	9186	36.647	9.000	34.795	75.889	15.279
OUT	9186	0.362	0.091	0.333	0.800	0.052
BOARD	9186	9.192	3	9	18	1.856
DUAL	9186	0.185	0	0	1	0.388
MKR	9186	0.784	0	1	1	0.412
Option	9186	0.054	0	0	1	0.227

Note:  $PL\_DIF$  is the range of TMT average pay in each industry, that is, the difference between the highest and lowest average pay.  $PL\_DIF$  and  $PAY\_GAP$  are all in ten thousands Yuan.

Table 3 Correlations between main variables.

	ROA	TOBINQ	GAP_1	GAP_2	PL	SOE
ROA	1					
TOBINQ	0.246***	1				
$GAP\_1$	0.217***	0.022**	1			
$GAP\_2$	0.027***	0.028***	0.597***	1		
PL	0.205***	$-0.030^{***}$	0.466***	0.004	1	
SOE	$-0.102^{***}$	$-0.105^{***}$	$-0.122^{***}$	$-0.212^{***}$	0.097***	1

<sup>\*</sup> Significance at the 0.1 level.

# 4. Empirical results

#### 4.1. Descriptive statistics

Table 2 presents descriptive statistics of the main variables in the 2005–2012 sample period. The mean (median) values of performance measures ROA and TOBINQ are 0.040 (0.037) and 1.727 (1.370), respectively, and those of  $GAP\_1$  and  $GAP\_2$  are 11.552 (11.594) and 1.642 (1.429), respectively. All of the variables are reasonably distributed without extreme observations. The unscaled values of TMT pay gap ( $PAY\_GAP$ ) indicate an average difference between CEO pay and median non-CEO pay of about 180,000 yuan. However, the difference varies widely across the sample, ranging from 5000 yuan to 1,500,000 yuan. The average PL value is 0.527, which means 52.7% of firms offer a TMT pay level that is higher than the median pay level in the same industry. The mean across industries in the range of average TMT pay in an industry ( $PL\_DIF$ ) is about 1,750,000 yuan, showing that even in the same industry, average TMT pay can differ widely. Table 2 also shows that 56.1% of the sample firms are SOEs, and that 18.5% of the firms have a CEO and chairperson who are the same individual (mean DUAL = 0.185). Further, 70% of firms operate in regions with a high rate of marketization, and only 5.4% grant stock options to their top executives (average Option value = 0.054).

Table 3 presents the correlation coefficients between the main variables. GAP is positively related to firm performance. The correlation coefficient between  $GAP\_1$  and  $GAP\_2$  is 0.597, significant at the 1% level,

<sup>\*\*</sup> Significance at the 0.05 level.

<sup>\*\*\*</sup> Significance at the 0.01 level.

Table 4 Univariate comparisons of SOE and non-SOE subsamples.

Variable	Non-SOE		SOE		Mean difference
	No. of observations	Mean	No. of observations	Mean	
Panel A					
CEOPAY	4031	518305.300	5155	522105.800	-3800.443
VPMPAY	4031	290198.800	5155	347235.200	$-57037.430^{***}$
PL	4031	0.472	5155	0.570	$-0.098^{***}$
$GAP\_1$	4031	11.707	5155	11.431	0.277***
$GAP\_2$	4031	1.800	5155	1.519	0.281***
ROA	4031	0.047	5155	0.035	0.012***
TOBINQ	4031	1.856	5155	1.626	0.230***
Option	4031	0.102	5155	0.017	0.086***
Variable	Non-SOE				Mean difference
	PL = 1		PL = 0	<del></del> ,	
	No. of observations	Mean	No. of observations	Mean	
Panel B					
$GAP\_1$	2129	11.168	1902	12.312	$-1.144^{***}$
$GAP\_2$	2129	1.736	1902	1.871	$-0.135^{***}$
Variable	SOE				Mean difference
	PL = 1		PL = 0		
	No. of observations	Mean	No. of observations	Mean	
Panel C					
$GAP\_1$	2219	10.840	2936	11.878	$-1.037^{***}$
$GAP\_2$	2219	1.546	2936	1.498	0.048***

Note: CEOPAY is the pay of the CEO, and VPMPAY is the median pay of non-CEOs.

which means that the two measures of pay gap are closely related. Further, TMT pay level PL is positively related to  $GAP\_1$  and uncorrelated with  $GAP\_2$ . SOE is negatively related to GAP.

Table 4 reports the results of univariate analysis for the different ownership subsamples. Panel A presents the comparison for the main variables. Average CEO pay does not differ between the SOE and non-SOE subsamples, although the average non-CEO pay of the non-SOE sample is significantly lower than that of the SOE sample. These results show clearly that the TMT pay gap is much greater in non-SOEs than in SOEs, and are also consistent with the view that TMTs in SOEs are more egalitarian. Further, SOE TMTs enjoy a higher pay level, but exhibit weaker firm performance, than their non-SOE counterparts. Panels B and C show the relationship between TMT pay gap and pay level in non-SOEs and SOEs, respectively. In non-SOEs, a higher TMT pay level is accompanied by a larger pay gap, whereas in SOEs, a higher TMT pay level is associated with a larger absolute pay gap ( $GAP_1$ ) but a smaller relative pay gap ( $GAP_2$ ).

# 4.2. Top management team pay level and pay gap

Table 5 presents the estimation results of model (1). Panels A and B show the regression results for accounting performance (*ROA*) and market performance (*TOBINQ*), respectively, with columns 4 and 8 reporting the IV regression results.<sup>12</sup> The results in column 1 of Panels A and B show that if TMT pay level (*PL*) is not controlled for, *GAP* is significantly positively related to firm performance at the 1% level, which is consistent

<sup>\*</sup> Significance at the 0.1 level.

<sup>\*\*</sup> Significance at the 0.05 level.

<sup>\*\*\*</sup> Significance at the 0.01 level.

<sup>&</sup>lt;sup>12</sup> We use the IVs stated above. We present only the second-stage IV regression results here.

Table 5
Relationship between firm performance and TMT pay level and TMT pay gap.

	ROA/GAP_1				ROA/GAP_2			
	OLS			IV	OLS			IV
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel A								
Constant	$-0.194^{***}$	$-0.129^{***}$	$-0.113^{***}$	$-0.235^{***}$	$-0.178^{***}$	$-0.105^{***}$	$-0.102^{***}$	$-0.161^{***}$
	(-9.613)	(-6.230)	(-5.074)	(-7.518)	(-8.758)	(-5.158)	(-4.960)	(-6.859)
GAP	0.00666***	0.00374***	0.00232***	0.00723***	0.000225	0.0000444	$-0.00202^*$	$-0.00556^{**}$
	(13.020)	(6.517)	(2.760)	(4.225)	(0.273)	(0.055)	(-1.654)	(-2.271)
PL		0.0146	-0.0166	-0.105		0.0182***	0.0118	-0.0142**
G (D Dr		(12.476)	(-1.338)	(-4.023)		(17.565)	(4.333)	(-2.173)
$GAP \times PL$			0.00272**	0.00750***			0.00388**	0.00976
COL	0.00000***	0.00722***	(2.531)	(3.333)	0.00760***	0.00027***	(2.449)	(2.599)
SOE	-0.00608***	$-0.00732^{***}$	$-0.00724^{***}$	$-0.00321^{**}$	-0.00768	$-0.00837^{***}$	-0.00824	-0.00733
LEV	(-4.833) $-0.120****$	(-5.799) $-0.118***$	(-5.736) $-0.118***$	(-2.205) $-0.120***$	(-5.956) $-0.123***$	(-6.547) $-0.119***$	(-6.451) $-0.119$ ***	(-5.217) $-0.119***$
LEV	-0.120 $(-29.418)$	-0.118 $(-28.960)$	-0.118 $(-29.067)$		-0.123 $(-30.038)$	-0.119 $(-29.138)$	-0.119 $(-29.160)$	-0.119 $(-35.232)$
SGROW	$0.0227^{***}$	$0.0225^{***}$	0.0226***	$(-34.225)$ $0.0235^{***}$	0.0230***	$0.0227^{***}$	0.0227***	$(-33.232)$ $0.0230^{***}$
SUKOW	(14.495)	(14.479)	(14.510)	(17.650)	(14.406)	(14.431)	(14.416)	(18.040)
SIZE	0.00877***	0.00721***	0.00712***	0.00991***	0.0109***	0.00778***	0.00774***	0.0104***
SIZL	(12.073)	(9.825)	(9.656)	(13.587)	(15.216)	(10.635)	(10.577)	(14.776)
TOP1	0.000209***	0.000211***	0.000215***	0.000222***	0.000172***	0.000195***	0.000198***	0.000186***
1011	(5.695)	(5.806)	(5.900)	(5.264)	(4.656)	(5.350)	(5.436)	(4.554)
CEOAge	0.00146	-0.000143	-0.0000292	0.00489	0.00464	0.000919	0.000846	0.00648
0201180	(0.370)	(-0.036)	(-0.007)	(1.030)	(1.158)	(0.234)	(0.215)	(1.407)
INDEPB	$-0.0328^{***}$	$-0.0350^{***}$	-0.0341***	$-0.0318^{***}$	$-0.0335^{***}$	$-0.0359^{***}$	$-0.0352^{***}$	$-0.0350^{***}$
	(-2.993)	(-3.214)	(-3.135)	(-2.629)	(-3.051)	(-3.297)	(-3.238)	(-3.002)
BOARD	0.000310	0.0000650	0.0000775	0.000457	0.000321	0.00000914	0.0000199	0.000193
	(1.016)	(0.215)	(0.256)	(1.221)	(1.041)	(0.030)	(0.066)	(0.534)
DUAL	$-0.00248^*$	-0.00188	-0.00196	$-0.00549^{***}$	-0.000844	-0.000986	-0.00100	-0.00216
	(-1.719)	(-1.318)	(-1.373)	(-3.463)	(-0.581)	(-0.690)	(-0.702)	(-1.416)
MKR	$0.00430^{***}$	$0.00270^{**}$	$0.00284^{**}$	$0.00642^{***}$	$0.00630^{***}$	$0.00323^{**}$	$0.00324^{**}$	0.00585***
	(3.144)	(1.987)	(2.087)	(4.189)	(4.572)	(2.364)	(2.380)	(3.941)
Option	0.0145***	$0.0130^{***}$	$0.0130^{***}$	0.0161***	$0.0150^{***}$	$0.0129^{***}$	0.0128***	0.0144***
	(8.068)	(7.272)	(7.234)	(6.538)	(8.259)	(7.153)	(7.104)	(6.046)
N	9186	9186	9186	6624	9186	9186	9186	6784
Adj. R-sq	0.284	0.295	0.296	0.234	0.270	0.292	0.292	0.268
	TOBINQ/GA	AP_1			TOBINQ/GA			
	OLS			IV	OLS			IV
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel B								
Constant	7.093***	7.886***	8.433***	7.240***	7.192***	7.920***	7.997***	7.979***
	(19.845)	(20.909)	(21.065)	(11.982)	(20.232)	(21.608)	(21.733)	(17.436)
GAP	0.0414***	0.00603	$-0.0422^{***}$	0.0549*	0.00778	0.00597	$-0.0368^*$	$-0.0880^{*}$
	(4.141)	(0.543)	(-2.684)	(1.658)	(0.518)	(0.402)	(-1.931)	(-1.843)
PL	, ,	0.177***	$-0.882^{***}$	$-1.866^{***}$	` ,	0.183***	0.0513	$-0.476^{***}$
		(8.073)	(-3.941)	(-3.702)		(9.309)	(1.039)	(-3.736)
$GAP \times PL$		` ′	0.0922***	0.134***		` /	0.0804***	0.232***
			(4.700)	(3.086)			(2.824)	(3.163)
SOE	0.0801***	$0.0650^{***}$	0.0679***	0.148***	0.0718***	0.0649***	0.0675***	0.118***
	(3.529)	(2.854)	(2.988)	(5.247)	(3.142)	(2.846)	(2.964)	(4.293)
LEV	$-0.387^{***}$	$-0.358^{***}$	$-0.366^{***}$	$-0.422^{***}$	$-0.406^{***}$	$-0.360^{***}$	$-0.361^{***}$	$-0.409^{***}$
	(-5.995)	(-5.536)	(-5.671)	(-6.217)	(-6.280)	(-5.539)	(-5.567)	(-6.189)
SGROW	0.0936***	0.0915***	0.0939***	0.109***	0.0955***	$0.0916^{***}$	0.0918***	0.0996***
	(3.372)	(3.342)	(3.429)	(4.236)	(3.447)	(3.349)	(3.352)	(4.008)

Table 5 (continued)

	ROA/GAP_1				ROA/GAP_2			
	OLS			IV	OLS	IV		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SIZE	-0.308***	-0.326***	$-0.329^{***}$	-0.295***	-0.295***	-0.326***	$-0.327^{***}$	-0.290***
	(-22.617)	(-23.201)	(-23.284)	(-20.889)	(-22.042)	(-23.049)	(-23.065)	(-21.120)
TOP1	$-0.00416^{***}$	$-0.00414^{***}$	$-0.00400^{***}$	$-0.00415^{***}$	$-0.00439^{***}$	$-0.00416^{***}$	$-0.00410^{***}$	$-0.00434^{***}$
	(-6.587)	(-6.543)	(-6.330)	(-5.080)	(-6.962)	(-6.589)	(-6.500)	(-5.461)
CEOAge	0.0910	0.0716	0.0754	0.112	0.110	0.0723	0.0708	0.0896
	(1.273)	(1.005)	(1.059)	(1.219)	(1.538)	(1.018)	(0.997)	(0.998)
INDEPB	0.259	0.232	0.264	0.260	0.254	0.231	0.244	0.241
	(1.157)	(1.041)	(1.188)	(1.111)	(1.137)	(1.035)	(1.093)	(1.059)
BOARD	0.00536	0.00239	0.00281	0.00570	0.00554	0.00240	0.00262	0.00382
	(1.040)	(0.465)	(0.547)	(0.786)	(1.074)	(0.467)	(0.511)	(0.542)
DUAL	$-0.0679^{**}$	$-0.0607^{**}$	-0.0633**	$-0.116^{***}$	$-0.0589^{**}$	$-0.0603^{**}$	$-0.0607^{**}$	$-0.0818^{***}$
	(-2.480)	(-2.223)	(-2.326)	(-3.785)	(-2.127)	(-2.194)	(-2.209)	(-2.748)
MKR	$-0.0582^{**}$	$-0.0776^{***}$	$-0.0731^{***}$	-0.0252	$-0.0458^*$	$-0.0768^{***}$	$-0.0764^{***}$	-0.0416
	(-2.427)	(-3.231)	(-3.048)	(-0.851)	(-1.916)	(-3.198)	(-3.182)	(-1.440)
Option	0.159***	0.141***	0.139***	0.191***	0.162***	0.141***	0.140***	0.180***
	(3.503)	(3.098)	(3.057)	(4.010)	(3.564)	(3.100)	(3.070)	(3.865)
N	9186	9186	9186	6624	9186	9186	9186	6784
Adj. R-sq	0.359	0.364	0.365	0.308	0.358	0.364	0.364	0.321

Note: Panel A shows the effects on *ROA*, and Panel B the effects on *TOBINQ*. We present both the ordinary least squares (OLS) estimation and IV estimation results, although we show only the second-stage results of the latter. The figures in parentheses are robust *t*-statistics adjusted for heteroskedasticity. We also control for year and industry fixed effects in all models.

with the majority of the Chinese literature. However, after *PL* is added to the model, the regression coefficients of *GAP* decrease, and the *R*-square of the models increases. For example, in column 2 of Panel A, the coefficient relating *GAP* to *ROA* decreases to 0.00374 from 0.00666 (in column 1), and the *R*-square increases from 28.4% to 29.5%. Similarly, in column 2 of Panel B, the coefficient's influence on *TOBINQ* declines from 0.0414 (significant at the 1% level) to 0.00603 (not significant at conventional levels), and the *R*-square rises from 35.9% to 36.4%. These results indicate that, in addition to TMT pay gap, TMT pay level is also positively related to firm performance.

The model in columns 2 and 6 do not consider that, in addition to its main effect, TMT pay level may also relate to firm performance through its interaction with TMT pay gap. The models in columns 3 and 7 of Panels A and B, allow such an interaction. The estimation results indicate that the coefficients of  $GAP \times PL$  are significantly positive at the 1% level, which suggests that the pay level can positively affect pay gap-induced incentives. These results are consistent with hypothesis H1. They indicate that the positive tournament effects of the pay gap are greater when the pay level is above the industry median.

In terms of economic significance, taking the sample's average total assets as an example, if a TMT's pay level is higher than that of its industry peers, then every 10,000-yuan increase in the pay gap raises net profits by about 42,600,000 yuan. If, in contrast, a TMT's pay level is lower than the industry average, every 10,000-yuan increase in the pay gap boosts net profits by about 19,800,000 yuan only. These results, which are economically significant, demonstrate that the tournament incentives induced by a pay gap are more pronounced when the TMT receives average pay that exceeds the industry median pay.

Columns 4 and 8 of Panels A and B present the results of the IV regressions. The coefficients of  $GAP \times PL$  are again significantly positive, which means that after controlling for the endogeneity of GAP, PL and  $GAP \times PL$ , TMT pay level still exerts a positive effect on pay gap efficiency, i.e., a higher pay level is more likely to induce positive pay gap incentives. In summary, Table 5 results are consistent with the prediction of H1.

<sup>\*</sup> Significance level of 0.1.

<sup>\*\*</sup> Significance level of 0.05.

<sup>\*\*\*</sup> Significance level of 0.01.

Table 6
Relationship between firm performance and TMT pay level and TMT pay gap for SOE and non-SOE subsamples.

IV regression	ROA				TOBINQ			
Independent	SOE	Non-SOE	SOE	Non-SOE	SOE	Non-SOE	SOE	Non-SOE
variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Constant	$-0.244^{***}$	-0.201***	$-0.176^{***}$	$-0.170^{***}$	7.122***	9.670***	7.987***	9.979***
	(-5.304)	(-4.015)	(-4.764)	(-3.953)	(8.327)	(8.521)	(12.121)	(11.814)
$GAP\_1$	0.00881***	$0.00368^*$			0.0789	0.0343		
	(3.539)	(1.659)			(1.571)	(0.626)		
$GAP\_1 \times PL$	0.00578	0.0115***			0.0114	0.280***		
	(1.587)	(3.389)			(0.176)	(3.274)		
$GAP_2$			-0.00249	$-0.00840^{***}$			0.00304	$-0.136^*$
			(-0.774)	(-2.738)			(0.040)	(-1.906)
$GAP\_2 \times PL$			0.00415	0.0150			-0.00140	0.354***
			(0.895)	(2.873)			(-0.013)	(2.884)
PL	$-0.0797^*$	$-0.153^{***}$	-0.00218	-0.0273***	-0.427	$-3.669^{***}$	-0.135	-0.713***
	(-1.874)	(-3.743)	(-0.272)	(-2.852)	(-0.570)	(-3.639)	(-0.776)	(-3.207)
LEV	$-0.126^{***}$	-0.113***	$-0.127^{***}$	-0.112***	-1.065***	0.0712	-1.034***	0.0929
	(-20.324)	(-14.524)	(-20.680)	(-15.013)	(-9.828)	(0.524)	(-9.814)	(0.706)
SGROW	0.0233***	0.0235	0.0226	0.0232***	0.149***	0.0649	0.137***	0.0584
	(8.285)	(8.673)	(7.960)	(8.885)	(2.893)	(1.443)	(2.768)	(1.366)
SIZE	$0.00890^{***}$	0.00983***	$0.00976^{***}$	$0.0108^{***}$	$-0.237^{***}$	$-0.387^{***}$	$-0.238^{***}$	$-0.372^{***}$
	(8.342)	(5.865)	(9.011)	(6.609)	(-11.124)	(-11.509)	(-11.416)	(-11.513)
TOP1	0.000168***	0.000271***	$0.000104^*$	$0.000287^{***}$	$-0.00243^{**}$	$-0.00799^{***}$	$-0.00276^{***}$	$-0.00746^{***}$
	(2.902)	(4.358)	(1.852)	(4.699)	(-2.288)	(-6.014)	(-2.652)	(-5.803)
CEOAge	0.0117	0.000480	0.0148**	0.00140	-0.0703	-0.0245	-0.0710	-0.0435
	(1.595)	(0.075)	(2.108)	(0.223)	(-0.566)	(-0.184)	(-0.602)	(-0.331)
INDEPB	$-0.0476^{***}$	-0.00572	$-0.0534^{***}$	-0.00433	-0.479	0.831*	$-0.516^*$	0.896**
	(-3.228)	(-0.277)	(-3.757)	(-0.212)	(-1.600)	(1.875)	(-1.772)	(2.043)
BOARD	0.000310	0.000944	-0.00000236	0.000902	-0.000224	0.0198	-0.00131	0.0206
	(0.758)	(1.258)	(-0.006)	(1.223)	(-0.032)	(1.437)	(-0.192)	(1.522)
DUAL	-0.00207	$-0.00556^{***}$	0.00124	-0.00286	-0.0159	$-0.0923^{**}$	0.0112	-0.0542
	(-0.745)	(-2.649)	(0.462)	(-1.420)	(-0.335)	(-2.190)	(0.235)	(-1.319)
MKR	0.00270	$0.0110^{***}$	0.00250	$0.00987^{***}$	0.0500	-0.0287	0.0326	-0.0543
	(1.330)	(3.850)	(1.258)	(3.513)	(1.428)	(-0.481)	(0.965)	(-0.912)
Option	0.0287***	0.0131***	0.0303***	0.0111***	$0.470^{***}$	0.191***	0.495***	0.171***
	(5.458)	(6.013)	(5.886)	(5.342)	(3.706)	(3.436)	(3.930)	(3.220)
N	3505	3119	3595	3189	3505	3119	3595	3189
Adj. R-sq	0.284	0.193	0.303	0.220	0.362	0.301	0.375	0.316

Note: The figures in parentheses are robust *t*-statistics adjusted for heteroskedasticity. We control for year and industry fixed effects in all models.

#### 4.3. Influence of ownership status

The results on the relationship between TMT pay gap and firm performance under different ownership conditions are presented in Table 6. To control for endogeneity, the table presents the IV regression results using only the IVs stated above. Columns 1–4 show the results for *ROA*, and columns 5–8 the results for *TOBINQ*.

Table 6 shows that the coefficients of  $GAP \times PL$  are significantly positive for the non-SOE subsample but are not significant for the SOE subsample. This result is consistent with H2, and suggests that the TMT pay level in non-SOEs is more likely to moderate the influence of a within-TMT pay gap on firm performance. However, the SOE TMT pay level has no effect on pay gap efficiency, possibly because the top executives of government-regulated SOEs are not sensitive to their external pay standing among their industry peers. Hence, an external compensation comparison does little to change competition and cooperation within SOE TMTs.

<sup>\*</sup> Significance level 0.1.

<sup>\*\*</sup> Significance level of 0.05.

<sup>\*\*\*</sup> Significance level of 0.01.

Table 7
Relationship between firm performance and TMT pay gap for the lowest and highest deciles of TMT pay level.

Constant         0.766°         0.234°         36.80°         6.308°         −0.00398         0.0408         18.90°         12           GAP_I         −0.0109         0.00456°         −0.192°         0.0414         −0.0052         (0.770)         (16.564)         (12           GAP_I         −0.0109°         0.00456°         −0.192°         0.0414         −0.000155         0.00293°         −0.0763°         0           GAP_2         −0.0773°         −0.00427         −0.257         0.00177         −0.0291°         0.00330         −0.0573         −0.0573           SOE         −0.0773°         −0.00427         −0.257         0.00177         −0.0291°         0.00330         −0.0573         −0.0753           LEV         −0.199°         −0.123°         −0.0728         −0.0224         −0.161°         −0.144°         −0.242°         −0.024°           CF,7100         (−7.586)         (−0.162)         (−0.063)         (−1.7107°         (−1.9758)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)         (−1.735)	Fixed-effects	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Constant         0.766°         0.234°         36.80°         6.308°         -0.00398         0.0408         18.90°         12           GAP_I         -0.0109         0.00456°         -0.192°         0.0414	Independent variables	ROA		TOBINQ		ROA		TOBINQ	
$ \begin{array}{c} (2.087) & (1.717) & (7.834) & (2.127) & (-0.052) & (0.770) & (16.564) & (12.564) \\ (-0.109) & (0.00456) & -0.192 & 0.00414 \\ (-1.920) & (2.392) & (-2.619) & (0.997) \\ \hline \\ GAP\_2 & & & & & & & & & & & & & & & & & & &$		$\overline{SPL} = 1$	SPL = 10	$\overline{SPL} = 1$	SPL = 10	PL = 0	PL = 1	PL = 0	PL = 1
$ \begin{array}{c} (2.087) & (1.717) & (7.834) & (2.127) & (-0.052) & (0.770) & (16.564) & (12.564) \\ (-0.0109) & (0.00456) & -0.192 & 0.00414 \\ (-1.920) & (2.392) & (-2.619) & (0.997) \\ \hline \\ GAP\_2 & & & & & & & & & & & & & & & & & & &$	Constant	0.760**	0.234*	36.80***	6.308**	-0.00398	0.0408	18.90***	12.39***
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(2.087)	(1.717)		(2.127)	(-0.052)	(0.770)	(16.564)	(12.662)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	GAP_1	$-0.0109^*$	0.00456**	$-0.192^{***}$	0.0414				
SOE		(-1.920)	(2.392)	(-2.619)	(0.997)				
SOE         -0.0773***         -0.00427         -0.257         0.00177         -0.0291***         0.00330         -0.0573         -0.0273           LEV         -0.199***         -0.123**         -0.0728         -0.0224         -0.161**         -0.144**         -0.242*         -0           LEV         -0.199***         -0.123**         -0.0728         -0.0224         -0.161**         -0.144**         -0.242*         -0           (-5.710)         (-7.586)         (-0.162)         (-0.063)         (-17.107)         (-17.578)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.735)         (-1.745)	GAP_2					-0.000155	0.00293**	$-0.0763^{***}$	0.0339
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$						(-0.080)	(2.113)	(-2.640)	(1.325)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	SOE	$-0.0773^{***}$	-0.00427	-0.257	0.00177	-0.0291***	0.00330	-0.0573	-0.188**
SGROW         (-5.710)         (-7.586)         (-0.162)         (-0.063)         (-17.107)         (-19.758)         (-1.735)         (-0.063)           SGROW         0.0200***         0.0285***         0.145**         0.0178         0.0197***         0.0179***         0.0196         0.06           SIZE         -0.0276**         -0.00998*         -1.331***         -0.338***         0.000960         0.00668         -0.837***         -0.5           C-2.227)         (-1.951)         (-8.323)         (-3.026)         (0.352)         (0.305)         (-20.684)         (-15           TOP1         0.00157**         -0.000426*         -0.0211**         0.00300         0.000374**         0.000324**         -0.00825**         -0.00           CEOAge         0.0215         (-1.745)         (-2.142)         (0.563)         (2.185)         (2.786)         (-3.245)         (-1           CEOAge         0.0215         0.0153         -1.137         0.451         0.00856         0.00690         0.0177         0.4           (0.349)         (0.746)         (-1.429)         (1.010)         (0.674)         (0.849)         (0.094)         (2           INDEPB         0.0118         0.0521         1.207         -0.698		(-3.800)	(-0.437)	(-0.978)	(0.008)	(-4.885)	(0.661)	(-0.650)	(-2.035)
SGROW         0.0200**         0.0285**         0.145**         0.0178*         0.0197**         0.0179**         0.0196*         0.06           SIZE         -0.0276**         -0.00998*         -1.331**         -0.338**         0.000960         0.000608*         -0.837**         -0.5           C-2.227)         (-1.951)         (-8.323)         (-3.026)         (0.352)         (0.305)         (-20.684)         (-1:           TOP1         0.00157**         -0.000426*         -0.211**         0.0030         0.00037**         0.000324**         -0.00825**         -0.00           CEOAge         0.0215         0.0153         -1.137         0.451         0.00856         0.00690         0.0177         0.4           (0.349)         (0.746)         (-1.429)         (1.010)         (0.674)         (0.849)         (0.094)         (2           INDEPB         0.0118         0.0521         1.207         -0.698         0.0362         -0.0103         0.235         (0           BOARD         -0.00689         0.00156         0.00164         0.00319         -0.000178         -0.000336         0.0180         0.00           DUAL         0.00624         -0.00539         -0.552*         -0.116         0.00213	LEV	$-0.199^{***}$	-0.123****	-0.0728	-0.0224	$-0.161^{***}$	$-0.144^{***}$	$-0.242^{*}$	-0.0959
SIZE		(-5.710)	(-7.586)	(-0.162)	(-0.063)	(-17.107)	(-19.758)	(-1.735)	(-0.710)
SIZE         -0.0276*         -0.00998*         -1.331*         -0.338**         0.000960         0.000608         -0.837*         -0.5           TOP1         0.00157*         -0.000426*         -0.0211*         0.00300         0.000374*         0.000324*         -0.00825*         -0.00           (2.045)         (-1.745)         (-2.142)         (0.563)         (2.185)         (2.786)         (-3.245)         (-0.00           CEOAge         0.0215         0.0153         -1.137         0.451         0.00856         0.00690         0.0177         0.4           (0.349)         (0.746)         (-1.429)         (1.010)         (0.674)         (0.849)         (0.094)         (2           INDEPB         0.0118         0.0521         1.207         -0.698         0.0362         -0.0103         0.235         0           BOARD         -0.00689         0.00156         0.00164         0.00319         -0.000336         0.0180         0.00           bulk         (-1.462)         (1.030)         (0.027)         (0.097)         (-0.169)         (-0.496)         (1.152)         (0           bulk         0.00624         -0.00539         -0.552*         -0.116         0.00213         -0.0630*         -0.	SGROW	0.0200***	0.0285***	0.145*	0.0178	0.0197***	0.0179***	0.0196	0.0666***
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(3.309)	(7.298)	(1.862)	(0.209)	(11.818)	(13.322)	(0.792)	(2.685)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	SIZE	$-0.0276^{**}$	$-0.00998^*$	-1.331****	-0.338***	0.000960	0.000608	$-0.837^{***}$	$-0.570^{***}$
$ \begin{array}{c} (2.045)  (-1.745)  (-2.142)  (0.563)  (2.185)  (2.786)  (-3.245)  (-0.204) \\ CEOAge  0.0215  0.0153  -1.137  0.451  0.00856  0.00690  0.0177  0.451 \\ (0.349)  (0.746)  (-1.429)  (1.010)  (0.674)  (0.849)  (0.094)  (2.006) \\ INDEPB  0.0118  0.0521  1.207  -0.698  0.0362  -0.0103  0.235  0.051 \\ (0.099)  (1.339)  (0.782)  (-0.823)  (1.364)  (-0.563)  (0.597)  (0.007) \\ (0.099)  (1.339)  (0.0782)  (-0.823)  (1.364)  (-0.563)  (0.597)  (0.007) \\ (-1.462)  (1.030)  (0.027)  (0.097)  (-0.169)  (-0.496)  (1.152)  (0.007) \\ (-1.462)  (1.030)  (0.027)  (0.097)  (-0.169)  (-0.496)  (1.152)  (0.007) \\ (0.346)  (-0.897)  (-2.372)  (-0.888)  (0.509)  (-2.090)  (-1.053)  (-0.007) \\ (0.277)  (-2.139)  (-0.533)  (-0.950)  (-0.570)  (-1.842)  (-0.595)  (-3.007) \\ (0.277)  (-2.139)  (-0.533)  (-0.950)  (-0.570)  (-1.842)  (-0.595)  (-3.007) \\ (0.324)  (-1.128)  (0.860)  (-0.299)  (-0.108)  (0.802)  (1.748)  (1.007) \\ (0.007)  (-0.108)  (0.007)  (-0.108)  (0.007) \\ (0.007)  (-0.108)  (0.007)  (-0.007) \\ (0.007)  (-0.108)  (0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007) \\ (0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.007)  (-0.$		(-2.227)	(-1.951)	(-8.323)	(-3.026)	(0.352)	(0.305)	(-20.684)	(-15.446)
$ \begin{array}{c} CEOAge \\ CEOAge \\ CEOAge \\ CO.349) \\ CO.349) \\ CO.746) \\ CO.747) \\ C$	TOP1	0.00157**	$-0.000426^*$	-0.0211**	0.00300	0.000374**	0.000324***	$-0.00825^{***}$	-0.000829
$ \begin{array}{c} CEOAge \\ CEOAge \\ CEOAge \\ CO.349) \\ CO.349) \\ CO.746) \\ CO.747) \\ C$		(2.045)	(-1.745)	(-2.142)	(0.563)	(2.185)	(2.786)	(-3.245)	(-0.386)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	CEOAge	0.0215	0.0153	-1.137	0.451	0.00856	0.00690	0.0177	0.440***
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.349)	(0.746)	(-1.429)	(1.010)	(0.674)	(0.849)	(0.094)	(2.932)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	INDEPB	0.0118		1.207	-0.698	0.0362	-0.0103	0.235	0.0876
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.099)	(1.339)	(0.782)	(-0.823)	(1.364)	(-0.563)	(0.597)	(0.259)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	BOARD	-0.00689	0.00156	0.00164	0.00319	-0.000178	-0.000336	0.0180	0.000991
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(-1.462)	(1.030)	(0.027)	(0.097)	(-0.169)	(-0.496)	(1.152)	(0.079)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	DUAL	0.00624	-0.00539	$-0.552^{**}$	-0.116	0.00213	$-0.00630^{**}$	-0.0653	-0.000542
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.346)	(-0.897)	(-2.372)	(-0.888)	(0.509)	(-2.090)	(-1.053)	(-0.010)
Option         0.0101         -0.00533         0.347         -0.0308         -0.000706         0.00225         0.169*         0.00225           (0.324)         (-1.128)         (0.860)         (-0.299)         (-0.108)         (0.802)         (1.748)         (1.748)           N         742         1035         742         1035         4348         4838         4348	MKR	0.0105	-0.0312**	-0.260	-0.303	-0.00519	$-0.0138^*$	-0.0804	-0.440***
N (0.324) (-1.128) (0.860) (-0.299) (-0.108) (0.802) (1.748) (1.748) (1.748) (1.748) (1.748) (1.748) (1.748) (1.748)		(0.277)	(-2.139)	(-0.533)	(-0.950)	(-0.570)	(-1.842)	(-0.595)	(-3.163)
N (0.324) (-1.128) (0.860) (-0.299) (-0.108) (0.802) (1.748) (1.748) (1.748) (1.748) (1.748) (1.748)	Option	0.0101	-0.00533	0.347	-0.0308	-0.000706	0.00225	0.169*	0.100*
N 742 1035 742 1035 4348 4838 4348	•	(0.324)	(-1.128)	(0.860)	(-0.299)	(-0.108)	(0.802)	(1.748)	(1.935)
A 1' D 0.405 0.241 0.140 0.004 0.215 0.142 0.100	N	` /	. ,		` /	` /	` /	` /	4838
Adj. $R$ -sq $-0.405$ $-0.241$ $0.140$ $-0.094$ $-0.215$ $-0.142$ $0.196$	Adj. R-sq	-0.405	-0.241	0.140	-0.094	-0.215	-0.142	0.196	0.153

Note: The figures in parentheses are robust t-statistics, and we also control for year and industry fixed effects in all models.

#### 5. Robustness tests

# 5.1. Subdivision of TMT pay level

Our main focus is on the effects of a TMT pay gap on firm performance under different TMT pay level conditions. As previously noted, firms often use their industry peers as a benchmark when agreeing contracts with top executives (Jiang, 2011). Accordingly, we divide the TMT pay level into two groups, one with a pay level higher than the industry median and the other with a pay level lower than the industry median, and find a stronger positive pay gap effect on firm performance when the TMT pay level is higher than the industry median.

In Table 5, the coefficients of the relative pay gap measure  $GAP_2$  are all significantly negative, meaning that when TMT members' pay level is lower (higher) than that of their industry peers,  $GAP_2$  is negatively (positively) related to firm performance. Hence, we can say that the median pay level in a given industry may be the flex point for the effect of a relative pay gap (i.e.,  $GAP_2$ ) on firm performance. However, the same cannot be said for the absolute pay gap measure (i.e.,  $GAP_1$ ).  $GAP_1$  is positively related to firm performance in both the higher and lower pay level groups, although more strongly so in the former. To ensure that

<sup>\*</sup> Significance level 0.1.

<sup>\*\*</sup> Significance level of 0.05.

<sup>\*\*\*</sup> Significance level of 0.01.

Table 8
Relationship between firm performance and TMT pay gap for the lowest and highest deciles of TMT pay level for SOEs and non-SOEs.

Independent variables	$\Delta ROA$				$\Delta$ TOBINQ			
	SOE	Non-SOE	SOE	Non-SOE	SOE	Non-SOE	SOE	Non-SOE
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Constant	$-0.0804^{**}$	-0.0323	$-0.0607^{**}$	-0.0541	4.053***	6.510***	4.173***	6.168***
	(-2.234)	(-0.751)	(-1.977)	(-1.406)	(8.363)	(9.607)	(8.781)	(9.964)
$GAP_{-1}_{t-1}$	0.00321**	-0.00168	,	` ′	0.00706	-0.0534**	, ,	,
	(2.146)	(-1.269)			(0.366)	(-2.397)		
$GAP_{-1_{t-1}} \times PL_t$	-0.00254	0.00341**			-0.0291	0.0682**		
	(-1.487)	(2.043)			(-1.251)	(2.413)		
$GAP_{-2_{t-1}}$	,	,	0.00279	$-0.00251^*$	,	, ,	0.0596	-0.0359
			(1.630)	(-1.752)			(1.519)	(-1.320)
$GAP_{-2_{t-1}} \times PL_t$			-0.00316	0.00262			$-0.0855^{*}$	0.0681
			(-1.408)	(1.403)			(-1.825)	(1.943)
PL	0.0363*	-0.0273	0.0143***	0.00781**	0.379	$-0.624^{*}$	0.174**	0.0282
	(1.891)	(-1.429)	(3.917)	(2.155)	(1.437)	(-1.894)	(2.414)	(0.421)
LEV	$-0.0679^{***}$	-0.0595***	$-0.0670^{***}$	-0.0603***	-0.354***	0.0984	-0.345***	0.104
	(-10.590)	(-7.824)	(-10.509)	(-7.933)	(-4.166)	(0.887)	(-4.216)	(0.945)
SGROW	0.0208***	0.0252***	0.0206***	0.0248***	-0.0210	-0.0571	-0.0255	-0.0614
	(7.778)	(10.013)	(7.956)	(10.181)	(-0.544)	(-1.213)	(-0.690)	(-1.351)
SIZE	0.00318***	0.00318**	0.00338***	0.00360**	-0.145***	-0.255***	-0.144***	$-0.257^{***}$
	(3.691)	(2.050)	(3.920)	(2.347)	(-8.935)	(-10.411)	(-9.305)	(-10.694)
TOP1	0.0000157	0.000129**	0.0000177	0.000146***	0.0000156	-0.00102	0.000154	-0.000765
	(0.331)	(2.465)	(0.379)	(2.795)	(0.018)	(-0.940)	(0.187)	(-0.710)
CEOAge	0.00693	0.00346	0.00818	0.00322	-0.0351	-0.0485	-0.0645	-0.0738
0201180	(1.115)	(0.591)	(1.335)	(0.563)	(-0.345)	(-0.483)	(-0.656)	(-0.738)
INDEPB	-0.0251**	-0.0100	-0.0258**	-0.00773	-0.295	0.271	-0.334	0.298
11,121112	(-2.079)	(-0.539)	(-2.171)	(-0.420)	(-1.267)	(0.866)	(-1.484)	(0.960)
BOARD	-0.0000926	-0.000317	-0.0000881	-0.000128	-0.00136	0.0129	0.000141	0.0122
20.1112	(-0.280)	(-0.488)	(-0.259)	(-0.198)	(-0.246)	(1.345)	(0.026)	(1.283)
DUAL	0.000697	-0.00371**	0.000673	$-0.00329^*$	0.0362	-0.00112	0.0276	-0.00871
	(0.283)	(-1.963)	(0.274)	(-1.755)	(0.992)	(-0.034)	(0.757)	(-0.271)
MKR	0.000559	0.00421	0.000598	0.00371	0.00880	-0.0463	0.00655	-0.0539
1,1111	(0.326)	(1.615)	(0.354)	(1.431)	(0.332)	(-1.026)	(0.254)	(-1.189)
Option	0.0174***	0.00661***	0.0179***	0.00580***	0.235***	0.0250	0.253***	0.0366
op.non	(4.659)	(3.898)	(4.833)	(3.410)	(2.592)	(0.662)	(2.769)	(0.980)
$ROA_{t-1}$	$-0.535^{***}$	$-0.608^{***}$	$-0.529^{***}$	$-0.610^{***}$	(2.372)	(0.002)	(2.70)	(0.500)
1011[-]	(-16.571)	(-16.573)	(-16.617)	(-16.874)				
$TOBINQ_{t-1}$	( 10.5/1)	(10.575)	( 10.017)	( 10.074)	$-0.372^{***}$	$-0.335^{***}$	$-0.370^{***}$	-0.341***
10DIIV2t-1					(-14.116)	(-13.165)	(-14.332)	(-13.441)
N	3505	3119	3595	3189	3477	2990	3566	3058
Adj. R-sq	0.317	0.350	0.315	0.356	0.528	0.502	0.530	0.503
110j. 11-5q	0.517	0.550	0.515	0.330	0.520	0.502	0.550	0.505

Note: The figures in parentheses are robust *t*-statistics adjusted by heteroskedasticity, and we also control for year and industry fixed effects in all models.

an absolute pay gap's influence on firm performance also experiences a flex point, we further subdivide TMT pay level. We first arrange it in ascending order for each year and each industry, and then divide the sample into 10 pay level groups. Firms in the first group (SPL=1) have the lowest pay level relative to their peers, i.e., lower than 90% of firms in the same industry, whereas those in the tenth group (SPL=10) have the highest such pay level, i.e., higher than 90% of firms in the same industry. We then estimate model (3) separately for each of the 10 groups to examine the absolute TMT pay gap's influence on firm performance. <sup>13</sup>

<sup>\*</sup> Significance at the 0.1 level.

<sup>\*\*</sup> Significance at the 0.05 level.

<sup>\*\*\*</sup> Significance at the 0.01 level.

The control variables in (3) are the same as those in model (1), and (3) also controls for fixed effects.

$$PERF_{t} = \beta_{0} + \beta_{1}GAP_{t} + \beta_{2}SOE_{t} + \beta_{3}LEV_{t} + \beta_{4}SGROW_{t} + \beta_{5}SIZE_{t} + \beta_{6}TOP1_{t} + \beta_{7}CEOAge_{t}$$

$$+ \beta_{8}BOARD_{t} + \beta_{9}INDEPB_{t} + \beta_{10}DUAL_{t} + \beta_{11}MKR_{t} + \beta_{12}Option_{t} + \sum Industry_{t}$$

$$+ \sum Year_{t} + \varepsilon_{t},$$

$$(3)$$

We also estimate model (3) for the PL = 0 and PL = 1 subsamples separately to examine the relative TMT pay gap's  $(GAP_2)$  influence on firm performance. The results are presented in Table 7.

Columns 1–4 of Table 7 show  $GAP\_I$ 's effects on firm performance in the highest and lowest pay level groups. In the lowest group (SPL=1),  $GAP\_I$  is negatively related to firm performance, whereas in the highest (SPL=10), it is positively related. These results are consistent with H1. Columns 5–8 show  $GAP\_2$ 's influence on firm performance in the higher- and lower-than-industry-median pay level groups. If a firm's pay level is lower (higher) than the median level,  $GAP\_2$  is negatively (positively) related to firm performance. These results are consistent with those in Table 5, and further suggest that the direction of a pay gap's influence on firm performance changes from negative to positive with an increase in pay level, which is consistent with H1.

# 5.2. Endogeneity of TMT pay level and TMT pay gap

In addition to the 2SLS regression approach with IVs, we use a model with change in performance ( $\Delta PERF$ ) as the dependent variable and lagged pay gap as the independent variable to control for endogeneity. A TMT pay gap in the previous year is less likely to be caused by performance growth in the current year, and the TMT pay level is less endogenous with performance growth than with current year performance. Hence, we adopt model (4) to examine the influence of the previous year's pay gap on performance growth under different ownership conditions.

$$\Delta PERF_{t} = \beta_{0} + \beta_{1}GAP_{t-1} + \beta_{2}PL_{t} + \beta_{3}GAP_{t-1} \times PL_{t} + \beta_{4}PERF_{t-1} + \beta_{5}SOE_{t} + \beta_{6}LEV_{t} + \beta_{7}SGROW_{t}$$

$$+ \beta_{8}SIZE_{t} + \beta_{9}TOP1_{t} + \beta_{10}CEOAge_{t} + \beta_{11}BOARD_{t} + \beta_{12}INDEPB_{t} + \beta_{13}DUAL_{t}$$

$$+ \beta_{14}MKR_{t} + \beta_{15}Option_{t} + \sum Industry_{t} + \sum Year_{t} + \varepsilon_{t}$$

$$(4)$$

The variables are defined as before, and we also control for the previous year's performance. The results of model (4) for the different ownership conditions are presented in Table 8. They show that the interaction term  $GAP_{t-1} \times PL_t$  is significantly positive for the non-SOE subsample and insignificantly negative for the SOE subsample. Hence, when the pay level of a TMT in a non-SOE is higher than the industry median, a larger TMT pay gap brings about more performance growth. For SOEs, in contrast, a higher pay level does not induce better TMT tournament incentives. When these firms' TMTs receive less pay than their peers, a within-TMT pay gap may be conducive to performance growth. <sup>14</sup>

The results are consistent if we remove observations for which the TMT's pay standing relative to industry peers changed. We also use a lagged pay level with pay gap to re-examine model (1) under different ownership conditions, and the results are consistent with those in Tables 6 and 7. All of these results provide further support for H2.

We also use several other methods to re-measure TMT pay level and find consistent results. To exclude the effects of high CEO pay, we re-define TMT pay level without the CEO's pay and find consistent results. Additionally, we also use only the top three executives' pay or excess TMT pay to measure the TMT pay level, and find consistent results.

# 6. Additional analyses

According to the descriptive data in Table 4, the TMT pay levels in SOEs are much higher than those in non-SOEs in the same industry. Many Chinese scholars have suggested that the excess pay levels of SOEs are not conducive to firm efficiency because top executives' pay is regulated by the government (Wu et al., 2010; Li

<sup>&</sup>lt;sup>14</sup> It is later suggested that the interaction's negative coefficients are all in central SOEs.

Table 9
Relationship between firm performance and TMT pay gap and TMT pay level for central SOEs and local SOEs: 2SLS-IV estimation.

IV regression	Central SO	Е			Local SOE			
Independent variables	ROA		TOBINQ		ROA		TOBINQ	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Constant	$-0.386^{***}$	$-0.209^{***}$	2.647	4.269***	$-0.259^{***}$	-0.198***	8.514***	8.874***
	(-3.575)	(-2.859)	(1.366)	(2.801)	(-5.411)	(-5.158)	(10.234)	(13.111)
$GAP\_1$	0.0174**		0.312**		0.0102***	, , , , ,	0.0571	· ·
_	(2.535)		(2.534)		(3.718)		(1.200)	
$GAP\_1 \times PL$	-0.00758		-0.339***		0.00902**		0.134**	
_	(-1.042)		(-2.598)		(2.521)		(2.154)	
$GAP\_2$	,	0.0171	,	1.295**	,	-0.00247	` /	0.0355
_		(0.775)		(2.371)		(-0.656)		(0.537)
$GAP\_2 \times PL$		$-0.016\hat{1}$		$-1.476^{**}$		0.00249		-0.0316
_		(-0.659)		(-2.446)		(0.337)		(-0.244)
PL	0.0677	0.0324	3.082**	1.740*	$-0.125^{***}$	-0.000466	$-1.798^{**}$	0.00833
	(0.862)	(0.861)	(2.186)	(1.866)	(-3.045)	(-0.040)	(-2.508)	(0.041)
LEV	$-0.107^{***}$	-0.108***	$-1.365^{***}$	$-1.417^{***}$	$-0.135^{***}$	-0.135***	-0.913***	$-0.916^{***}$
	(-11.623)	(-12.204)	(-8.244)	(-8.177)	(-22.553)	(-23.441)	(-8.744)	(-9.069)
SGROW	0.0227***	0.0210***	0.0666	0.0462	0.0229***	0.0221***	0.182***	0.165***
	(6.732)	(6.506)	(1.102)	(0.739)	(10.277)	(10.481)	(4.684)	(4.439)
SIZE	0.00698***	0.00606***	$-0.126^{***}$	-0.143***	0.0108***	0.0119***	-0.288***	-0.274***
	(3.800)	(3.608)	(-3.822)	(-4.399)	(9.477)	(10.858)	(-14.528)	(-14.230)
TOP1	0.0000327	0.0000241	-0.00939***	-0.0102***	0.000203***	0.000110*	-0.0000616	-0.00106
	(0.307)	(0.228)	(-4.914)	(-4.992)	(3.007)	(1.734)	(-0.052)	(-0.950)
CEOAge	0.0310**	0.0295**	-0.0458	0.00991	0.00234	0.00976	-0.131	$-0.14\hat{1}$
0	(2.252)	(2.235)	(-0.185)	(0.039)	(0.276)	(1.196)	(-0.890)	(-0.979)
INDEPB	-0.0402	$-0.0484^{*}$	-0.544	-0.562	-0.0503***	$-0.0496^{***}$	-0.515	$-0.580^{*}$
	(-1.364)	(-1.736)	(-1.029)	(-1.033)	(-2.598)	(-2.684)	(-1.529)	(-1.781)
BOARD	-0.000634	-0.000748	$-0.0025\hat{1}$	-0.0104	0.000978*	0.000379	0.00235	-0.000317
	(-0.750)	(-0.913)	(-0.166)	(-0.655)	(1.770)	(0.725)	(0.245)	(-0.035)
DUAL	0.000292	0.000615	-0.00862	-0.0851	-0.00402	0.000841	-0.00846	0.0413
	(0.049)	(0.103)	(-0.080)	(-0.722)	(-1.332)	(0.291)	(-0.161)	(0.814)
Option	0.00885	0.0110	-0.0404	0.00398	0.0388***	0.0392***	0.667***	0.683***
•	(0.822)	(1.052)	(-0.209)	(0.020)	(5.893)	(6.149)	(5.816)	(6.100)
MKR	0.00282	0.00169	0.222***	0.137*	0.00199	0.000422	0.0110	-0.00937
	(0.724)	(0.427)	(3.172)	(1.740)	(0.865)	(0.192)	(0.275)	(-0.242)
N	1012	1038	1012	1012	2493	2557	2493	2557
Adj. R-sq	0.231	0.270	0.347	0.308	0.283	0.332	0.359	0.378

Note: The figures in parentheses are robust t-statistics, and we also control for year and industry in all models.

et al., 2014). We find that excess pay levels in SOEs also do not influence internal TMT competition or cooperation, and thus should be addressed as the next step in China's ongoing comprehensive SOE reform. On 29 August 2014, the country's Central Political Bureau passed the *Pay System Reform Program of Central Management Companies' Responsible Person*, which was formally implemented on 1 January 2015. The program requires that the first 72 central SOE top executives receive a pay cut, thereby setting an example for local SOE pay reform. However, the program does not provide the detailed rules on how the pay cuts should be carried out or stipulate whether the internal pay distribution should be considered. Moreover, it is also important to discuss what notifications local SOEs should be given when they consider central SOEs' approach to top executive compensation reform.

As previously noted, if a company's pay incentive system is effective, the TMT pay level will moderate the TMT pay gap's efficiency. Thus, firms should consider choosing a suitable pay gap to ensure a certain pay standing in the industry. Because SOEs are regulated by the government, their industry pay standing is not

<sup>\*</sup> Significance level of 0.1.

<sup>\*\*</sup> Significance level of 0.05.

<sup>\*\*\*</sup> Significance level of 0.01.

related to the market. Hence, cutting their top managers' pay should not directly hurt the efficiency of an internal pay gap. However, this is not true for all SOEs because, in China, central and local SOEs are subject to different constraints (Xia and Fang, 2005), different levels of government intervention (Pan et al., 2008), and different pay systems. Although they both belong to the state-owned system, their top managers' preferences concerning monetary compensation differ, leading to different degrees of sensitivity to the firm's external pay standing. Thus, we re-estimate model (1) separately for central SOEs and local SOEs. The results are presented in Table 9. For consistency, we present only the results of the 2SLS regression with the IVs stated above.

Columns 1–4 of Table 9 report the results for central SOEs. The effect of a TMT pay gap on accounting performance (ROA) does not differ in the different pay level groups, although the pay gap's effect on market performance (TOBINQ) is stronger when the pay level is lower, which suggests that reducing the compensation of top managers in central SOEs is actually beneficial to firm value. When these SOEs implement pay cutting measures, their internal TMT pay gap could be raised.

Columns 5–8 of Table 9 present the results for local SOEs, showing that a higher TMT pay level is beneficial to the incentives induced by an absolute pay gap, but has little effect on those induced by a relative pay gap. These results suggest that in a local SOE, the TMT's pay level influences within-team competition and cooperation to some degree. Thus, when local SOEs engage in top manager compensation reform, direct pay cuts may not be the best approach. For those with a high pay level, increasing the internal pay gap may be a good solution to boost efficiency.

In summary, our results suggest that the pay cutting policy currently being implemented in China's SOEs will not damage the internal efficiency of a central SOE TMT, but local SOEs should not blindly follow the pay cutting decisions of their central counterparts. Because the pay level in local SOE TMTs can also affect within-team pay gap efficiency, the redistribution of payments within the TMT may be the optimal solution. In addition, we find the TMT pay level in local SOEs to have modulating effects in regions with a higher marketization level and in more competitive industries. We do not tabulate the results of these regressions because of space limitations.

#### 7. Conclusion

Although the incentive effects of a TMT pay gap have received considerable research attention, the results of prior research are conflicting. These conflicting results are consistent with the two dominant theories in this area, namely tournament theory and social comparison theory, which make conflicting predictions. One explanation for the conflicting findings is that each of these theories holds, but only under specific conditions. TMT pay level concerns an external comparison and TMT pay gap an internal comparison. Most of the literature examine the efficiency of these two components of the pay system separately, with few studies noting that an external comparison also influences the internal comparison. Moreover, firms often use the market pay level in their industry as a benchmark in designing their own TMT pay contracts.

The mandatory requirement that Chinese listed firms provide information on their top executives' compensation, makes it possible to conduct external comparisons and to also study whether these external comparisons have implications for the managerial incentive and firm performance effects of internal comparisons. Using compensation data from 2005 to 2012, we examine the effects of TMT pay level on within-team pay gap efficiency. Our results show that a higher TMT pay level than that of the team's industry peers stimulates within-team competence, and thus indicates that tournament theory may be more suitable than social comparison theory for explaining the incentives of a pay gap. A lower TMT pay level than industry peers, in contrast, renders top executives more sensitive to an internal pay gap because they perceive their pay to be unfair. Accordingly, social comparison theory provides a more suitable explanation under these circumstances. Therefore, when firms are designing contracts for top executives, it is important that they both adjust the pay level to suit the market and ensure an appropriate within-TMT pay distribution, which is vital to the pay system as a whole. For example, the Chinese listed firm *Everfine Photo* (stock ID: 300306) emphasizes in its 2012 annual report that "the firm has implemented [a] new reward system, [and] the top managers' pay ... keep[s] up with the market, and should make dynamic maintenance of the pay gap and pay level every year according to the market level...."

However, because of government regulation, providing SOE top managers with excess compensation is not only unfavorable to firm performance (Wu et al., 2010), but also fails to stimulate internal TMT competition. As noted earlier, a reform of the pay system for central SOEs has been in effect since 1 January 2015, and consequently the majority of these firms' top executives will receive a pay cut. Our findings suggest that such pay cuts will not adversely affect the incentives induced by an internal pay gap and may even boost firm value, although the same does not hold true for local SOEs. The top managers of central and local SOEs face different degrees of government intervention, market competition and other factors, with the former group's pay system being more market-oriented. Directly cutting pay is likely to be detrimental to the incentives induced by an internal TMT pay gap in local SOEs. For high-paying local SOEs, widening the internal pay gap may be a solution to boosting efficiency.

# References

Akerlof, G.A., Yellen, J.L., 1986. Efficiency Wage Models of the Labor Market. Cambridge University Press, Cambridge.

Bloom, M., 1999. The performance effects of pay dispersion on individuals and organizations. Acad. Manage. J. 42 (1), 25-40.

Bloom, M., Michel, J.G., 2002. The relationships among organizational context, pay dispersion, and managerial turnover. Acad. Manage. J. 45 (1), 33–42.

Bognanno, M.L., 2001. Corporate tournaments. J. Labor Econ. 19 (2), 290-315.

Brown, M., 2001. Unequal pay, unequal responses? Pay referents and their implications for pay level satisfaction. J. Manage. Stud. 38, 879–896.

Brown, M.P., Sturman, M.C., Simmering, M.J., 2003. Compensation policy and organizational performance: the efficiency, operational, and financial implications of pay levels and pay structure. Acad. Manage. J. 46 (6), 752–762.

Chen, D., Chen, X., Wan, H., 2005. Compensation regulation and job consumption in state-owned enterprises. Econ. Res. J. 2, 100–111 (in Chinese).

Chen, J., Ezzamel, M., Cai, Z., 2011. Managerial power theory, tournament theory, and executive pay in china. J. Corp. Finan. 17 (4), 1176–1199.

Chen, Z., Zhang, M., 2006. Research on top executives' pay distributions. China Account. Rev. 1, 15-28 (in Chinese).

Colquitt, J.A., Conlon, D.E., Wesson, M.J., Porter, C.O., Ng, K.Y., 2001. Justice at the millennium: a meta-analytic review of 25 years of organizational justice research. J. Appl. Psychol. 86 (3), 425–445.

Core, J.E., Holthausen, R.W., Larcker, D.F., 1999. Corporate governance, chief executive officer compensation, and firm performance. J. Financ. Econ. 51 (3), 371–406.

Cowherd, D.M., Levine, D.I., 1992. Product quality and pay equity between lower-level employees and top management: an investigation of distributive justice theory. Adm. Sci. Q. 37, 302–320.

Deutsch, M., 1985. Distributive Justice: A Social-Psychological Perspective. Yale University Press, New Haven, CT.

Drago, R., Garvey, G.T., 1998. Incentives for helping on the job: theory and evidence. J. Labor Econ. 16 (1), 1-25.

Ehrenberg, R.G., Bognanno, M.L., 1990a. Do tournaments have incentive effects? J. Polit. Econ. 98 (6), 1307-1324.

Ehrenberg, R.G., Bognanno, M.L., 1990b. The incentive effects of tournaments revisited: evidence from the European PGA tour. Ind. Labor Relat. Rev. 43 (3), 74S-88S.

Eriksson, T., 1999. Executive compensation and tournament theory: empirical tests on Danish data. J. Labor Econ. 17 (2), 262-280.

Fan, G., Wang, X.R., 2007. Marketization Index in China: A Report on the Process of Marketization in Different Regions. Economic Science Press (in Chinese).

Fang, J.X., 2012. Top executives' excess compensation and corporate governance policy. Manage. World 11, 144-155 (in Chinese).

Frank, R.H., 1985. Choosing the Right Pond: Human Behavior and the Quest for Status. Oxford University Press, New York.

Fredrickson, J.W., Davis-Blake, A., Sanders, W.M., 2010. Sharing the wealth: social comparisons and pay dispersion in the CEO's top team. Strateg. Manage. J. 31 (10), 1031–1053.

Gerhart, B., Milkovich, G.T., 1992. Employee compensation: research and practice. In: Dunnette, M.D., Hough, L.M. (Eds.), Handbook of Industrial and Organizational Psychology, vol. 3. Consulting Psychologists Press, Palo Alto, CA, pp. 481–569.

Hambrick, D., Siegel, P., 1997. Compensation patterns within top management groups: on the harmful effects of executive pay disparities in high-technology firms. Organ. Sci. 16 (3), 259–274.

Harbring, C., Irlenbusch, B., 2003. An experimental study on tournament design. Labour Econ. 10 (4), 443-464.

He, W., Hao, P., 2014. Does departmental pay dispersion really deteriorate employee affective commitment? The moderators of pay level, pay-for-performance intensity and workforce diversity. Nankai Bus. Rev. 4, 13–23 (in Chinese).

Hibbs Jr., D.A., Locking, H., 2000. Wage dispersion and productive efficiency: evidence for Sweden. J. Labor Econ. 18 (4), 755–782. Hills, F.S., 1980. The relevant other in pay comparisons. Ind. Relat. 19, 345–351.

Jensen, M.C., 1986. Agency costs of free cash flow, corporate finance and takeovers. Am. Econ. Rev. 76 (2), 323-329.

Jiang, W., 2011. Marketization level, industry competition and the growth of managerial compensation. Nankai Bus. Rev. 5, 58–67 (in Chinese).

Kale, J.R., Reis, E., Venkateswaran, A., 2009. Rank-order tournaments and incentive alignment: the effect on firm performance. J. Financ. 64 (3), 1479–1512.

- Kepes, S., Delery, J., Gupta, N., 2009. Contingencies in the effects of pay range on organizational effectiveness. Pers. Psychol. 62 (3), 497–531
- Kini, O., Williams, R., 2012. Tournament incentives, firm risk, and corporate policies. J. Financ. Econ. 103 (2), 350-376.
- Knoeber, C.R., Thurman, W.N., 1994. Testing the theory of tournaments: an empirical analysis of broiler production. J. Labor Econ. 12 (2), 155–179.
- Lambert, R.A., Larcker, D.F., Weigelt, K., 1993. The structure of organizational incentives. Adm. Sci. Q. 38, 438-461.
- Law, K.S., Wong, C., 1998. Relative importance of referents on pay satisfaction: a review and test of a new policy-capturing approach. J. Occup. Organ. Psychol. 71, 47–60.
- Lazear, E.P., 1989. Pay equality and industrial politics. J. Polit. Econ. 97, 561-580.
- Lazear, E.P., Rosen, S., 1981. Rank-order tournaments as optimum labor contracts. J. Polit. Econ. 89 (5), 841-864.
- Li, W., Cen, Y., Hu, Y., 2014. Does external pay gap encourage top management? An empirical study based on managerial market and ownership type. Nankai Bus. Rev. 4, 24–35 (in Chinese).
- Li, W., Hu, Y., 2012. Who is encouraged by pay dispersion in state-owned enterprises? Econ. Res. 12, 125-136 (in Chinese).
- Liao, L., Liao, G., Shen, H., 2009. Operating risk, promotion incentive and corporate performance. China Ind. Econ. 8, 119–130 (in Chinese).
- Lin, B., Lu, R., 2009. Managerial power, compensation gap and firm performance—evidence from Chinese public listed companies. Global Financ. J. 20 (2), 153–164.
- Lin, J., Huang, Z., Sun, Y., 2003. TMT pay gap, firm performance and corporate governance. Econ. Res. 4, 31–41 (in Chinese).
- Liu, C., Sun, L., 2010. A study on relation of salary difference and firm performance: evidence from state-owned enterprises. Nankai Bus. Rev. 2, 40–51 (in Chinese).
- Lu, Z., Wang, X., Zhang, P., 2012. Do Chinese state-owned enterprises pay higher wages? Econ. Res. 3, 28-39 (in Chinese).
- Main, B.G., O'Reilly III, C.A., Wade, J., 1993. Top executive pay: tournament or teamwork? J. Labor Econ. 11 (4), 606-628.
- Messersmith, J.G., Guthrie, J.P., Ji, Y., Lee, J., 2011. Executive turnover: the influence of dispersion and other pay system characteristics. J. Appl. Psychol. 96 (3), 457–469.
- Milkovich, G.T., Newman, J.M., 2002. Compensation. Irwin, Chicago, IL.
- Myers, S.C., 1977. Determinants of corporate borrowing. J. Financ. Econ. 5 (2), 147-175.
- Oldman, G.R., Kulik, C.T., Stepina, L.P., Ambrose, M.L., 1986. Relations between situational factors and comparative referents used by employees. Acad. Manage. J. 29, 599–608.
- O'Reilly III, C.A., Main, B.G., Crystal, G.S., 1988. CEO compensation as tournament and social comparison: a tale of two theories. Adm. Sci. Q. 33 (2), 257–274.
- Pan, H., Xia, X., Yu, M., 2008. Government intervention, political connections and the mergers of local government-controlled enterprises. Econ. Res. 4, 41–52 (in Chinese).
- Pfeffer, J., Langton, N., 1993. The effect of wage dispersion on satisfaction, productivity, and working collaboratively: evidence from college and university faculty. Adm. Sci. Q. 38, 382–407.
- Qi, H., Zou, Y., 2014. The empirical research on incentive effect of the external fairness of executive compensation on agents' behavior. Account. Res. 3, 26–32 (in Chinese).
- Rosen, S., 1986. Prizes and incentives in elimination tournaments. Am. Econ. Rev. 76 (4), 701-715.
- Shaw, J.D., Gupta, N., Delery, J.E., 2002. Pay dispersion and workforce performance: moderating effects of incentives and interdependence. Strateg. Manage. J. 23 (6), 491–512.
- Siegel, P.A., Hambrick, D.C., 2005. Pay disparities within top management groups: evidence of harmful effects on performance of high-technology firms. Organ. Sci. 16 (3), 259–274.
- Trevor, C.O., Wazeter, D.L., 2006. A contingent view of reactions to objective pay conditions: interdependence among pay structure characteristics and pay relative to internal and external referents. J. Appl. Psychol. 91, 1260–1275.
- Wu, L., Lin, J., Wang, Y., 2010. The external fairness of CEO's emolument, the nature of stock rights and company performance. Manage. World 3, 117–126 (in Chinese).
- Xia, L., Fang, Y., 2005. Government control, institutional environment and firm value: evidence from the Chinese securities market. Econ. Res. 5, 40–51 (in Chinese).
- Xu, L., Xin, Y., Chen, G., 2006. The types of controlling shareholders and the performance of China's listed companies. World Econ. 10, 78–89 (in Chinese).
- Yang, R., Wang, Y., Nie, H., 2013. The promotion mechanism of "quasi-official": evidence from Chinese central state-owned enterprises. Manage. World 3, 23–33 (in Chinese).
- Zhang, Z.T., 2007. Top management team coordination needs, compensation dispersion and firm performance: a perspective of tournament theory. Nankai Bus. Rev. 2, 4–11 (in Chinese).
- Zhang, Z.T., 2008. An empirical study on the impact of the internal pay gap on future performance. Account. Res. 9, 81–87 (in Chinese).
- Zhang, Z., Wang, Y., Liu, N., 2012. Study on the design factors and modes of team compensation scheme. Econ. Manage. 8, 89–96 (in Chinese).
- Zhou, Q., Zhu, W., 2010. On the incentive effects of SOE tournament. China Econ. Q. 2, 5715–5796 (in Chinese).